ORIGINAL RESEARCH

Anti-Osteoporosis Is Imperative in Prevention of Progress of Ankylosing Spondylitis

Bin Zheng*, Panfeng Yu*, Haiying Liu, Yan Liang

Spine Surgery, Peking University People's Hospital, Beijing, People's Republic of China

*These authors contributed equally to this work

Correspondence: Yan Liang, Email liangyan_spine@163.com

Background: Ankylosing spondylitis (AS) is a chronic autoimmune disease that affects the spine and peripheral joints, often leading to kyphosis, joint stiffness, and even ankylosis. Sagittal parameters, such as total thoracic kyphosis (TTK), thoracic kyphosis (TK), major thoracic kyphosis (MTK), and thoracolumbar kyphosis (TLK), are crucial indices for evaluating spinal alignment in AS patients and can reflect disease progression. This study aims to explore the relationship between bone mineral density (BMD), sagittal parameters, and joint ankylosis in AS patients.

Methods: A retrospective study was conducted on 147 AS patients. Participants were divided into three groups based on cervical and hip joint mobility. BMD was measured using quantitative computed tomography (QCT). Sagittal parameters (TTK, TK, MTK, TLK) were assessed using X-rays. Ordinal multinomial logistic regression and Spearman correlation analyses were performed to identify factors influencing joint stiffness.

Results: Significant differences in age, BMD, and sagittal parameters (TTK, TK, MTK, TLK) were observed among the groups. Ordinal logistic revealed that BMD (Estimate = 0.012) was negatively correlated with joint stiffness, while TTK (Estimate = 0.020) and TLK (Estimate = 0.030) were positively correlated. Age, TK, and MTK do not have a significant impact on joint stiffness. Spearman analysis showed no correlation between BMD and sagittal parameters (TTK and TLK). Besides, TTK and TLK were correlated.

Conclusion: In AS patients, BMD is an independent protective factor against joint stiffness, whereas sagittal parameters (TTK and TLK) contribute to increased joint stiffness. These findings highlight the importance of monitoring both bone mineral density and key sagittal parameters in clinical practice. Early anti-osteoporosis treatment, alongside interventions targeting abnormal spinal alignment, may help preserve joint mobility and potentially prevent progression to joint ankylosis.

Plain Language Summary: BMD is independent protective factor for joint ankylosis in ankylosis spondylitis. It is imperative to start early anti-osteoporosis treatment to preserve cervical and hip mobility.

Keywords: ankylosis spondylitis, BMD, osteoporosis, kyphosis, sagittal parameters

Introduction

Ankylosing spondylitis (AS) is a chronic progressive autoimmune disease primarily affecting the sacroiliac joints, spine processes, paraspinal soft tissues, and peripheral joints. It can also present with extra-articular manifestations. Clinically, it mainly manifests as pain in the lower back, neck, buttocks, hips, and swelling of joints.^{1–3} In severe cases, spine deformities, cervical ankylosis, and hip joint ankylosis can occur.^{4–6} The primary sites of inflammation are the attachments of ligaments and joint capsules. In the early stages, there is local congestion, edema, and infiltration of inflammatory cells, leading to the formation of granulation tissue, which quickly undergoes fibrosis and ossification. Secondary ossification and the formation of new bone result in bone sclerosis and joint ankylosis.⁷

291

Osteoporosis is the most common bone disease, characterized by low bone mass and microarchitectural deterioration of bone tissue, leading to increased bone fragility and susceptibility to fractures.⁸ Dual-energy X-ray absorptiometry (DXA) is the most recognized and accepted method for measuring bone mineral density (BMD).^{9,10} Quantitative computed tomography (QCT) is another method for measuring BMD, based on clinical CT scan data analyzed using QCT software. Due to the technical advantages of QCT and the rapid development of CT technology, QCT is gaining increasing attention in the field of osteoporosis research and clinical application globally.^{11–14}

Bone is a target in ankylosing spondylitis, characterized by extensive changes in bone remodeling due to the interaction between inflammation and bone. Combined analysis of DXA and QCT in patients with early and long-term disease has shown low BMD and bone loss in the spine and hips of AS patients.^{15–17} Normal cervical spine and hip joints, rich in spongy trabecular bone, are lightweight and act more like springs than levers, promoting flexibility rather than stiffness. Based on this pathological process, cervical spine and hip joint mobility are affected during the course of AS, potentially leading to ankylosis, which is a significant complication.¹⁸

Previous research has primarily focused on kyphosis and BMD in ankylosing spondylitis. There is currently no comprehensive study on the characteristics and relationship between BMD and joint ankylosis in patients with ankylosis ing spondylitis. This paper conducts a retrospective analysis of patients with AS who also have hip or cervical ankylosis to further explore the relationship between BMD and joint ankylosis.

Study Design

This study is a retrospective study investigating 147 AS patients from January 2016 to December 2019 in Peking University People's Hospital. Participants were divided into three groups based on cervical and hip joint mobility: Group A, with AS and both neck and hip joint ankylosis; Group B, with AS and either neck or hip joint mobility; and Group C, with AS but normal neck and hip joint mobility. This study has been approved by the Ethics Committee of Peking University People's Hospital (Ethics Approval Number: 2018PHC076), and all participants provided written informed consent.

Inclusion criteria were as follows: (1) diagnosed with ankylosing spondylitis; (2) preoperative X-rays and magnetic resonance imaging (MRI) of the spine and sacroiliac joints; (3) complete data for rheumatologist evaluation. Exclusion criteria included: (1) individuals with a history of spinal surgery; (2) individuals with spinal tumors, infections, or fractures; (3) individuals with incomplete patient information.

Data Collection

Data were obtained by reviewing hospital records and assessing disease status by rheumatologists and spine surgeons. Primary measurement indicators included patient symptoms, neck and hip joint mobility, MRI of the sacroiliac joints, BMD measured by QCT, and full-length anteroposterior and lateral X-rays of the spine.

Parameter Measurement

X-ray evaluations included comprehensive anteroposterior (AP) and lateral X-rays of the entire spine as the basis for morphological analysis., (1) Total Thoracic kyphosis (TTK): Sagittal angle between T1 and T12 (2) Thoracic kyphosis-(TK):Sagittal angle between T5 and T12 (3) Major Thoracic kyphosis(MTK): The angle between the proximal endplate of the vertebra above and the distal endplate of the vertebra below in the segment affected by kyphosis. (4) Thoracolumbar kyphosis(TLK) was determined by the angle between the superior endplate of the tenth thoracic vertebra (T10) and the inferior endplate of the second lumbar vertebra (L2).

Statistical Analysis

SPSS 29.0 software was used for analysis. The study was divided into three groups based on the presence of hip and neck ankylosis, with joint ankylosis as the dependent variable in an ordinal multinomial logistic regression analysis to explore the related influencing factors of joint stiffness in patients. Descriptive analysis was first used to describe the basic characteristics of all potential factors affecting joint stiffness in AS patients. Continuous and categorical variables between the three groups were analyzed using ANOVA. Subsequently, parameters with

Results

Basic Characteristics of Study Patients

A total of 147 patients were included in this study, with 19 patients diagnosed AS with both cervical and hip joint ankylosis, 84 patients with AS and either cervical or hip joint ankylosis, and 44 patients with AS without cervical or hip joint ankylosis. Table 1 records the basic conditions of the three groups. ANOVA results indicate no differences in gender, but significant differences in age, BMD, TTK, TK, MTK, and TLK among the groups (P<0.05). The AS+ cervical or hip ankylosis group has the highest mean age (43.26 ± 9.66), while the AS-only group is the youngest (36 ± 13). The AS-only group has the highest QCT value (89.39 ± 34.14). AS+cervical and hip ankylosis group has the highest values for TTK, TK, MTK, and TLK.

Results of Ordinal Multinomial Logistic Regression Analysis

Age, BMD, TTK, TK, MTK, and TLK were subjected to ordinal multinomial logistic regression, with the regression results shown in Table 2. The results indicate that BMD, TTK and TLK are related to joint stiffness in AS patients. BMD

	Kyphosis+Cervical and Hip Ankylosis	Kyphosis+Cervical or Hip Ankylosis	Kyphosis	Р
Age	40.1±8.19	43.26±9.66	36±13	P=0.001
Gender				P=0.772
Male	18	76	34	
Female	1	8	10	
QCT	70.53±21.84	76.06±33.28	89.39±34.14	P=0.041
ттк	91.28±41.13	63.94±25.52	46.46±24.63	P<0.001
тк	63.33±23.67	60.3±20.03	47.80±23.88	P=0.004
МТК	62.17±24.87	49.26±18.98	37.45±16.97	P<0.001
TLK	49.74±20.87	36.62±15.06	23.22±20.35	P<0.001

 Table I Characteristic of Three Groups

Table 2 Results of Ordinal Logistic Regression

	Estimate	Std. Error	Wald	df	Р	95% CI	
Age	0.029	0.017	2.88	I	0.09	-0.004	0.062
BMD	-0.012	0.006	4.34	I	0.04	-0.023	-0.001
ттк	0.02	0.008	6.56	I	0.01	0.005	0.035
тк	-0.009	0.013	0.47	I	0.495	-0.035	0.007
МТК	0.028	0.015	3.39	I	0.066	-0.002	0.057
LTK	0.03	0.011	7.51	I	0.006	0.009	0.051

Table	3	Spearman	Correlation	Analysis
Betwee	n R	elevant Fact	ors	

	QCT	ттк	TLK
QCT	-	-	-
ттк	-0.052	-	-
TLK	-0.029	0.497*	-

Note: *P<0.05, there is a correlation between the two parameters.

is negatively correlated with joint stiffness (Estimate=-0.012), while TTK (Estimate=0.02) and TLK (Estimate=0.03) are positively correlated with joint stiffness.

Spearman Analysis Results

To further clarify the intrinsic influencing factors between BMD, TTK and TLK, Spearman analysis was conducted, with the results shown in Table 3. It was found that BMD is not related to TTK and TLK, while TTK and TLK are mutually correlated.

Discussion

Inflammation in ankylosing spondylitis involves small joints, leading to bone erosion, joint space narrowing, and ankylosis. The initial stage of small joint ankylosis is the fusion of cartilage surfaces, progressing to bony ankylosis. Small joint ankylosis impairs normal joint function and affects joint mobility, resulting in joint stiffness.^{4,5} Figure 1 demonstrates a typical case of AS with hip joint and cervical ankylosis.

Previous research has primarily focused on the relationship between BMD and fractures in ankylosing spondylitis,^{19–}²¹ with no studies on the relationship between joint stiffness and BMD. This study uses QCT to evaluate BMD. Unlike DXA, QCT provides volumetric measurements of BMD and can differentiate between cortical and trabecular bone. Since syndesmophytes or bridging ankylosis may be present in AS, changes in cortical and trabecular bone may differ.^{22,23}

Descriptive studies have found that QCT-measured BMD is lower in AS patients compared to normal individuals. Moreover, patients with joint stiffness have lower QCT-measured BMD than those with only AS, meeting the current diagnostic criteria for osteoporosis. Inflammatory cytokines, glucocorticoid therapy, immobilization, and reduced physical activity due to joint pain and muscle weakness are considered major risk factors for low BMD in these diseases.^{24,25}

As the disease progresses, patients frequently develop significant sagittal spinal imbalance, including increased TK, decreased LL. Measuring and analyzing sagittal parameters (spinopelvic parameters) can provide a better assessment of the degree of deformity caused by ankylosing spondylitis, guide clinical decision-making, and predict prognosis.



Figure 1 AS patients with neck and hip ankylosis.

The results of the ordinal multinomial logistic regression analysis indicate that BMD, TTK and TLK are factors affecting joint stiffness in AS patients. BMD is negatively correlated with joint stiffness, while TTK and TLK are positively correlated. These results suggest that higher BMD have a protective effect, while TTK and TLK are risk factors for increased joint stiffness. These findings provide important clinical treatment references, suggesting that changes in these indicators should be closely monitored in managing AS patients. Further Spearman correlation analysis shows no significant correlation between BMD and sagittal parameters (TTK and TLK), while TTK and TLK are significantly correlated with each other. This result suggests that BMD independently influences joint stiffness. And TTK and TLK also significantly influence joint stiffness, interacting with each other and further affecting joint ankylosis.

Changes in bone mineral density and sagittal parameters in AS patients are not necessarily directly related. While both impact the degree of joint stiffness, their underlying mechanisms are distinct. I. Changes in bone mineral density mainly reflect a "systemic" or "broad regional" pathophysiological process. Decreased BMD often indicates overall bone loss or a risk of osteoporosis, which can be influenced by immune inflammation, endocrine hormones, and various other factors, as well as the patient's level of physical activity and nutritional status. Although patients with AS frequently exhibit reduced bone mass, the rate and pattern of bone loss may not be consistent across different spinal segments or joints.²⁶ II.Changes in sagittal parameters mainly reflect "localized or structural" alterations: TTK, TLK and other parameters measure the thoracic or thoracolumbar kyphosis angles, indicating changes in vertebral alignment as new bone (syndesmophytes, bridging) forms around the sacroiliac and facet joints. In AS, bone overgrowth (osteophytes, syndesmophytes) and bone loss (osteopenia or osteoporosis) can coexist in the same patient, yet they may not overlap in distribution. Some segments may exhibit marked osteophyte formation and structural fusion, while others have considerable bone loss.^{27–29} Hence, "loss of bone mineral density" and "increased thoracic or thoracolumbar kyphosis" do not necessarily share a straightforward one-to-one relationship. Although both affect the ultimate degree of joint stiffness, their underlying mechanisms may not be linearly interlinked.

Osteoporosis is a common complication in ankylosing spondylitis. This study suggests that BMD has a protective effect against joint stiffness. The relationship between AS and osteoporosis may involve the following mechanisms: 1. Ankylosing spondylitis is a chronic inflammatory disease that primarily affects the spine and sacroiliac joints. Persistent inflammation keeps the immune system continuously activated. Pro-inflammatory cytokines stimulate the production of RANKL, a key factor in osteoclast differentiation and activation. Osteoclasts are responsible for bone resorption. Enhanced osteoclast activity leads to accelerated bone resorption, where bone tissue is broken down faster than it is replaced.^{30,31} This imbalance in bone remodeling leads to localized osteoporosis, especially near the inflamed joints. Bone density decreases and becomes more porous. Osteoporosis itself further exacerbates bone resorption. As bones become more fragile, microfractures occur more easily, triggering local inflammation and further osteoclast activation.³² While ankylosing spondylitis causes bone loss through osteoporosis, it also leads to new bone formation (ankylosis) in certain areas. Fragile bones are more prone to this abnormal bone growth, leading to joint ankylosis. The combination of osteoporosis and ankylosis significantly increases the risk of ankylosis in patients with AS, especially in the later stages of the disease. Therefore, treatment targeting the inflammation of ankylosing spondylitis (such as TNF inhibitors) may help reduce osteoclast activity.^{33,34} Additionally, specific treatments for osteoporosis may be needed to prevent further bone loss, thereby slowing the progression of joint ankylosis. 2. AS patients often have limited mobility due to pain and inflammation, leading to osteoporosis due to prolonged lack of exercise. Lack of weight-bearing exercises and physical activities is a known risk factor for osteoporosis, a condition characterized by low bone mass and the deterioration of bone tissue.^{35,36} This makes the bones fragile and more susceptible to fractures. As osteoporosis progresses, it can cause additional pain, especially in the spine and hips. This new source of pain further hinders physical activity, creating a vicious cycle: inactivity leads to more bone loss, which in turn leads to more pain and further reduces activity. Pain caused by osteoporosis further limits patient activity, exacerbating cervical and other joint stiffness. Lack of exercise not only affects bone density but also exacerbates joint stiffness, particularly in the cervical spine and other joints affected by AS. This increased stiffness can further restrict the patient's range of motion and overall quality of life. Appropriate exercise can improve BMD, maintain bone structure, and reduce the risk of falls and joint mobility.³⁷

This study has several limitations. As a cross-sectional study, we did not assess whether BMD progression predicts the worsening of functional impairment. Longer prospective longitudinal studies are needed to examine the relationship

between joint stiffness and BMD in AS. Additionally, we did not quantify joint mobility limitations, making it difficult to grade joint abnormalities. Finally, although multiple indicators were included in this study, other potential factors, such as disease duration, medication use, or lifestyle factors, affecting joint stiffness were not included in the analysis. Future research should consider more comprehensive factors.

Conclusion

In AS patients, BMD is an independent protective factor against joint stiffness, whereas sagittal parameters (TLK and TTK) contribute to increased joint stiffness. These findings highlight the importance of monitoring both bone mineral density and key sagittal parameters in clinical practice. Early anti-osteoporosis treatment, alongside interventions targeting abnormal spinal alignment, may help preserve joint mobility and potentially prevent progression to joint ankylosis.

Ethics Approval

All procedures performed in studies involving human participants were in accordance with the 1964 helsinki declaration and its later amendments or comparable ethical standards, and this study has been approved by the Ethics Committee of Peking University People's Hospital (Ethics Approval Number: 2018PHC076).

Funding

This work was supported by the fund of Peking University People's Hospital(RDY2021-09), the fund of Peking University People's Hospital (2023HQ05).and the fund of National Orthopedics and Sports Rehabilitation Clinical Research Center, cultivation project "Application and promotion of pressure sensor assisted craniopelvic spinal traction device" (2021-NCRC-CXJJ-PY-38.

Disclosure

Bin Zheng, Panfeng Yu, Haiying Liu and Yan Liang declare no conflicts of interest.

References

- 1. Hanson A, Brown MA. Genetics and the causes of ankylosing spondylitis. *Rheum Dis Clin N Am.* 2017;43(3):401-414. doi:10.1016/j. rdc.2017.04.006
- 2. Mauro D, Thomas R, Guggino G, Lories R, Brown MA, Ciccia F. Ankylosing spondylitis: an autoimmune or autoinflammatory disease? *Nat Rev Rheumatol*. 2021;17(7):387–404. doi:10.1038/s41584-021-00625-y
- 3. Sieper J, Poddubnyy D. Axial spondyloarthritis. Lancet. 2017;390(10089):73-84. doi:10.1016/S0140-6736(16)31591-4
- Jung JY, Kim MY, Hong YS, Park SH, Kang KY. Association between facet joint ankylosis and functional impairment in patients with radiographic axial spondyloarthritis. Semin Arthritis Rheum. 2021;51(5):1005–1010. doi:10.1016/j.semarthrit.2021.07.015
- 5. Lee BW, Jung JY, Kim MY, Hong YS, Park SH, Kang KY. Prevalence and associated factors of facet joint ankylosis in patients with axial spondyloarthritis. *J Rheumatol*. 2023;50(6):763–768. doi:10.3899/jrheum.220749
- 6. Stal R, Sepriano A, van Gaalen FA, et al. Associations between syndesmophytes and facet joint ankylosis in radiographic axial spondyloarthritis patients on low-dose CT over 2 years. *Rheumatology*. 2022;61(12):4722–4730. doi:10.1093/rheumatology/keac176
- 7. Schett G. Bone formation versus bone resorption in ankylosing spondylitis. Mol Mech Spondyloarthropathies. 2009:114-121.
- 8. Compston JE, McClung MR, Leslie WD. Osteoporosis. Lancet. 2019;393(10169):364-376. doi:10.1016/S0140-6736(18)32112-3
- 9. Lorente-Ramos R, Azpeitia-Armán J, Muñoz-Hernández A, García-Gómez JM, Díez-Martínez P, Grande-Bárez M. Dual-energy x-ray absorptiometry in the diagnosis of osteoporosis: a practical guide. *AJR Am J Roentgenol*. 2011;196(4):897–904. doi:10.2214/AJR.10.5416
- 10. Messina C, Monaco CG, Ulivieri FM, Sardanelli F, Sconfienza LM. Dual-energy X-ray absorptiometry body composition in patients with secondary osteoporosis. *Eur J Radiol.* 2016;85(8):1493–1498. doi:10.1016/j.ejrad.2016.03.018
- 11. Abdalbary M, Sobh M, Elnagar S, et al. Management of osteoporosis in patients with chronic kidney disease. Osteoporos Int. 2022;33 (11):2259-2274. doi:10.1007/s00198-022-06462-3
- 12. Deshpande N, Hadi MS, Lillard JC, et al. Alternatives to DEXA for the assessment of bone density: a systematic review of the literature and future recommendations. *J Neurosurg Spine*. 2023;38(4):436–445. doi:10.3171/2022.11.SPINE22875
- 13. Engelke K. Quantitative computed tomography-current status and new developments. J Clin Densitom. 2017;20(3):309-321. doi:10.1016/j. jocd.2017.06.017
- 14. Lang TF. Quantitative computed tomography. Radiol Clin North Am. 2010;48(3):589-600. doi:10.1016/j.rcl.2010.03.001
- 15. Byrdy-Daca M, Duczkowski M, Sudoł-Szopińska I, et al. Spinal involvement in patients with chronic non-bacterial osteomyelitis (CNO): an analysis of distinctive imaging features. J Clin Med. 2023;12(23). doi:10.3390/jcm12237419
- Will R, Palmer R, Bhalla AK, Ring F, Calin A. Osteoporosis in early ankylosing spondylitis: a primary pathological event? *Lancet*. 1989;2 (8678-8679):1483-1485. doi:10.1016/S0140-6736(89)92932-2

- 17. Roux C. Osteoporosis in inflammatory joint diseases. Osteoporosis Int. 2011;22:421-433. doi:10.1007/s00198-010-1319-x
- 18. Galbusera F, Wilke H-J. Biomechanics of the Spine: Basic Concepts, Spinal Disorders and Treatments. Academic Press; 2018.
- Lukasiewicz AM, Bohl DD, Varthi AG, et al. Spinal fracture in patients with ankylosing spondylitis: cohort definition, distribution of injuries, and hospital outcomes. Spine. 2016;41(3):191–196. doi:10.1097/BRS.00000000001190
- 20. Pray C, Feroz NI, Nigil Haroon N. Bone mineral density and fracture risk in ankylosing spondylitis: a meta-analysis. *Calcif Tissue Int*. 2017;101 (2):182–192. doi:10.1007/s00223-017-0274-3
- Yan F, Wu L, Lang J, Huang Z. Bone density and fracture risk factors in ankylosing spondylitis: a meta-analysis. Osteoporos Int. 2024;35(1):25–40. doi:10.1007/s00198-023-06925-1
- 22. Klingberg E, Lorentzon M, Göthlin J, et al. Bone microarchitecture in ankylosing spondylitis and the association with bone mineral density, fractures, and syndesmophytes. *Arthritis Res Therapy*. 2013;15:1–11. doi:10.1186/ar4368
- 23. Martel D, Monga A, Chang G. Osteoporosis imaging. Radiol Clin North Am. 2022;60(4):537-545. doi:10.1016/j.rcl.2022.02.003
- Vestergaard P, Hermann P, Jensen JE, Eiken P, Mosekilde L. Effects of paracetamol, non-steroidal anti-inflammatory drugs, acetylsalicylic acid, and opioids on bone mineral density and risk of fracture: results of the Danish Osteoporosis Prevention Study (DOPS). Osteoporos Int. 2012;23 (4):1255–1265. doi:10.1007/s00198-011-1692-0
- 25. Canalis E, Mazziotti G, Giustina A, Bilezikian JP. Glucocorticoid-induced osteoporosis: pathophysiology and therapy. Osteoporos Int. 2007;18 (10):1319–1328. doi:10.1007/s00198-007-0394-0
- 26. Klingberg E, Lorentzon M, Mellström D, et al. Osteoporosis in ankylosing spondylitis-prevalence, risk factors and methods of assessment. *Arthritis Res Therapy*. 2012;14:1–12. doi:10.1186/ar3833
- Lee JS, Youn MS, Shin JK, Goh TS, Kang SS. Relationship between cervical sagittal alignment and quality of life in ankylosing spondylitis. *Eur Spine J.* 2015;24(6):1199–1203. doi:10.1007/s00586-014-3491-8
- 28. Son SM, Choi SH, Shin JK, Goh TS, Lee JS. Radiologic parameters of ankylosing spondylitis patients treated with anti-TNF-α versus nonsteroidal anti-inflammatory drugs and sulfasalazine. *Eur Spine J.* 2019;28(4):649–657. doi:10.1007/s00586-019-05912-7
- 29. Zhang YP, Qian BP, Qiu Y, et al. Sagittal vertical axias, spinosacral angle, spinopelvic angle, and T1 pelvic angle: which parameters may effectively predict the quality of life in ankylosing spondylitis patients with thoracolumbar kyphosis? *Clin Spine Surg.* 2017;30(7):E871–e6. doi:10.1097/BSD.00000000000463
- 30. Seong S, Kim JH, Kim N. Pro-inflammatory cytokines modulating osteoclast differentiation and function. J Rheum Dis. 2016;23(3):148–153. doi:10.4078/jrd.2016.23.3.148
- 31. Souza PP, Lerner UH. The role of cytokines in inflammatory bone loss. Immunol invest. 2013;42(7):555-622. doi:10.3109/08820139.2013.822766
- 32. Adejuyigbe B, Kallini J, Chiou D, Kallini JR. Osteoporosis: molecular pathology, diagnostics, and therapeutics. *Int J mol Sci.* 2023;24(19):14583. doi:10.3390/ijms241914583
- Perpétuo IP, Raposeiro R, Caetano-Lopes J, et al. Effect of tumor necrosis factor inhibitor therapy on osteoclasts precursors in ankylosing spondylitis. PLoS One. 2015;10(12):e0144655. doi:10.1371/journal.pone.0144655
- 34. Lata M, Hettinghouse AS, Liu C. Targeting tumor necrosis factor receptors in ankylosing spondylitis. Ann N Y Acad Sci. 2019;1442(1):5–16. doi:10.1111/nyas.13933
- 35. Castrogiovanni P, Trovato FM, Szychlinska MA, Nsir H, Imbesi R, Musumeci G. The importance of physical activity in osteoporosis. From the molecular pathways to the clinical evidence. 2016.
- 36. Benedetti MG, Furlini G, Zati A, Letizia Mauro G. The effectiveness of physical exercise on bone density in osteoporotic patients. *Biomed Res Int.* 2018;2018(1):4840531. doi:10.1155/2018/4840531
- 37. Dent E, Daly RM, Hoogendijk EO, Scott D. Exercise to prevent and manage frailty and fragility fractures. *Curr Osteoporos Rep.* 2023;21 (2):205–215. doi:10.1007/s11914-023-00777-8

International Journal of General Medicine



Publish your work in this journal

The International Journal of General Medicine is an international, peer-reviewed open-access journal that focuses on general and internal medicine, pathogenesis, epidemiology, diagnosis, monitoring and treatment protocols. The journal is characterized by the rapid reporting of reviews, original research and clinical studies across all disease areas. The manuscript management system is completely online and includes a very quick and fair peer-review system, which is all easy to use. Visit http://www.dovepress.com/testimonials.php to read real quotes from published authors.

Submit your manuscript here: https://www.dovepress.com/international-journal-of-general-medicine-journal

🖪 🛛 in 🗖

297